Author’s response to reviews

Title: Anti-inflammatory effect of Vaccinium oldhamii stems through inhibition of NF-κB and MAPK/ATF2 signaling activation in LPS-stimulated RAW264.7 cells

Authors:

Ha Na Kim (01083326745@nate.com)
Jeung Kyu Baek (bjk1099@naver.com)
Su Bin Park (govlqlsl@naver.com)
Jeong Dong Kim (shot5613@naver.com)
Ho Jun Son (shj7400@korea.kr)
Gwang Hun Park (ppkh0230@korea.kr)
Hyun Ji Eo (ehj56@naver.com)
Jae Ho Park (parkjh@jwu.ac.kr)
Hyuk Sang Jung (jhs@khu.ac.kr)
Jin Boo Jeong (jjb0403@anu.ac.kr)

Version: 2 Date: 11 Oct 2019

Author’s response to reviews:

1. Page 5, heading "Methods" should be replaced with "Materials and methods"
RESPONSE: We replaced "Methods" with "Materials and methods" accordingly.

2. In Materials and methods section, "Analysis of extracts" section should be placed after section of "Preparation of extracts". Accordingly Results section should be rearranged.
RESPONSE: It was revised accordingly.

3. Authors have mentioned that marker compounds analytical such as (+)-catechin, (-)-epicatechin, proanthocyanidin A2 and cinnamtannin were used in HPLC analysis. But the figure doesn't show single mentioned standard. Authors should provide HPLC chromatogram showing peaks of these compounds and the extract.
RESPONSE: It was revised accordingly.